Author’s response to reviews

Title: Maternal and Paternal Perinatal Depressive Symptoms Associate with 2- and 3-Year-Old Children’s Behaviour: Findings from the APrON Longitudinal Study

Authors:
Nicole Letourneau (nicole.letourneau@ucalgary.ca)
Brenda Leung (brenda.leung@uleth.ca)
Henry Ntanda (henry.ntanda@ucalgary.ca)
Deborah Dewey (dmdewey@ucalgary.ca)
Andrea Deane (andrea.deane@albertahealthservices.ca)
Gerald Giesbrecht (ggiesbre@ucalgary.ca)

Version: 3 Date: 03 Oct 2019

Author’s response to reviews:

1. Reviewer comment: Please ensure that the email address for the corresponding author matches that which has been listed on our editorial system - currently they differ.

   Author response: We would like to use Dr. Nicole Letourneau’s e-mail address (nicole.letourneau@ucalgary.ca) as the corresponding author's e-mail address within the manuscript. We're wondering if this manuscript would be able to be transferred to Dr. Letourneau's existing BMC Pediatrics account under this e-mail address. As one of her research assistants has created a separate account with BMC Pediatrics for the manuscript submission, we're unable to change the associated e-mail address for this account to nicole.letourneau@ucalgary.ca

2. Reviewer comment: As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

   Author response: Within this section, we declared that Dr. Nicole Letourneau is an Associate Editor with BMC Pediatrics.

3. Reviewer comment: Thank you for providing a Summary of Revisions. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.
Author response: We removed the Summary of Revisions from the submission's supplementary files.

4. Reviewer comment: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Author response: We uploaded the manuscript as a single, final, clean version.